Author's response to reviews

Title: Co-morbidity and drug treatment in Alzheimer's disease. A cross sectional study of participants in the Dementia Study in Northern Norway.

Authors:

Fred Andersen (frea@online.no)
Matti Viitanen (matti.viitanen@ki.se)
Dag S Halvorsen (dag.seeger.halvorsen@unn.no)
Bjorn Straume (bjorn.straume@ism.uit.no)
Torgeir A Engstad (torgeir.a.engstad@unn.no)

Version: 6 Date: 30 September 2011

Author's response to reviews: see over
Dear Editor

I have revised the manuscript according to your request

1. The trial registration number is removed.
2. The Method section is revised. The last sentence of the Participant paragraph clarifies that the study uses AD participants from the original RCT.
3. The Competing interest statement is revised.
4. Neither the study protocol nor the main results from the Dementia Study in Northern Norway have so far been published anywhere.
5. The present paper is based on data already collected in the main study. This is clarified in the Approval section by now.

I hope that the present statement and revision is in accordance with your expectations.

Kind regards

Fred Andersen